Author's response to reviews

Title: Near-infrared (NIR) spectroscopy. A new method for arthroscopic evaluation of low grade degenerated cartilage lesions. Results of a pilot study

Authors:

Gunter Spahn (spahn@pk-eisenach.de)
Holger Plettenberg (hplettenberg@fzmb.de)
Enriko Kahl (e.kahl@med.uni-goettingen.de)
Hans M Klinger (michael.klinger@med.uni-goettingen.de)
Thomas Muckley (Thomas.Mueckley@med.uni-jena.de)
Gunther O Hofmann (gunther.hofmann@med.uni-jena.de)

Version: 3 Date: 22 May 2007

Author's response to reviews: see over
Dear Biomed Central Editorial team,

attached you find the manuscript of our paper “Authors: Gunter Spahn, Holger Plettenberg, Enriko Kahl, Hans M Klinger, Thomas Muckley and Gunther O Hofmann Title : Near-infrared (NIR) spectroscopy. A new method for arthroscopic evaluation of low grade degenerated cartilage lesions. Results of a pilot study
Journal: BMC Musculoskeletal Disorders
MS : 1843489925130711”

Accordingly to your mail, the corrections were made as suggested:

Line numbers ∧ Please remove these. → done

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system. → done

Running title, Keywords ∧ Please remove these. → done

Introduction - Please rename this section as ‘Background’. → done

Materials & Methods - Please rename this section ‘Methods’. → done

Highlighting/Tracking ∧ Please remove any highlighting/tracking from your manuscript text. → done

Legend of Figures-> Figure legends → done

Authors’ contributions -> Authors’ contributions → done

Manuscript sections should include (in the following order): Abstract; Background; Methods; Results; Discussion; Conclusions; Abbreviations (if any); Competing interests; Authors’ contributions; Acknowledgements; References; Figure legends (if any); Tables (if any); Description of Additional files (if any). → done

Authors’ contributions - Please indicate that all authors read and approved the final manuscript. → done

References - The reference list should contain all authors’ names, if there are 30 or below. The term ‘et al.’ should not be used. → done, for formatting the reference list “Reference Manager” was used.

Best regards

G. Spahn